Author's response to reviews

Title: Prevalence of Congenital Heart Defect in Guangdong Province, 2008-2012

Authors:

Li Wu (wulistart@163.com)
Bing Li (522276188@qq.com)
Jianhong Xia (1140951465@qq.com)
Cunwei Ji (jwstart@163.com)
Zhijiang Liang (183558283@qq.com)
Yuanzhu Ma (506695391@qq.com)
Shiyun Li (190659606@qq.com)
Yuntao Wu (1500755513@qq.com)
Youjie Wang (wangyoujie@mails.tjmu.edu.cn)
Qingguo Zhao (1693910767@qq.com)

Version: 5 Date: 8 February 2014

Author's response to reviews:

Dear editor,

We appreciate your thoughtful, constructive comments, from which the manuscript has benefited greatly. We have carefully considered the comments and revised the manuscript accordingly. We hope that we have satisfactorily incorporated all the comments and suggestions. The following changes have been introduced regarding your comments.

Change the following statement in your manuscript: "We get written consent from the Birth defects Intervention Group of Guangdong Province to use the data." to -- "We received permission from the Birth defects Intervention Group of Guangdong Province to use the data."

#Response: Thanks for the editor for this point. We have changed the sentence "We get written consent from the Birth defects Intervention Group of Guangdong Province to use the data." to -- "We received permission from the Birth defects Intervention Group of Guangdong Province to use the data." in the manuscript.

Correspondence and phone calls about the paper should be directed to Qingguo Zhao at the following address, phone and fax number, and E-mail address:

Guangdong Women and Children Hospital, Xingnan Road 521, Guangzhou 510010, China, PR China
Phone: +86 20 39151605 Fax: +86 20 39151833
E-mail: 1693910767@qq.com

Thanks very much for your attention to our paper#

Sincerely yours#